Author's response to reviews

Title: A Cross-sectional study of stand-alone percutaneous coronary intervention in a Nigerian Cardiac Catheterization Laboratory

Authors:

Adeyemi Johnson (yemi@firstcardiologyconsultants.com)
Bode Falase (bodefalase@gmail.com)
Ifeoluwa Ajose (ifeajose@gmail.com)
Yemi Onabowale (onabowale@hotmail.com)

Version: 2 Date: 17 October 2013

Author's response to reviews: see over
The Editor  
BMC Cardiovascular disorders  

17th October 2013  

Dear Sir,  

UPDATE TO MANUSCRIPT: A CROSS-SECTIONAL STUDY OF STAND-ALONE PERCUTANEOUS CORONARY INTERVENTION IN A NIGERIA CARDIAC CATHETERIZATION LABORATORY  

As instructed, we have added an acknowledgments section to the above manuscript and have resubmitted the manuscript.  

Yours truly  

Bode Falase FWACS CTh, FRCS CTh